Author's response to reviews

Title: Population pharmacokinetic and pharmacodynamic modeling of transformed binary effect data of triflusal in healthy Korean male volunteers: a randomized, open-label, multiple dose, crossover study

Authors:

Sung Min Park (psm7271@naver.com)
Joomi Lee (joomibbc@hanmail.net)
Sook-Jin Seong (jy-nee@hanmail.net)
Jong Gwang Park (freesstyler@hanmail.net)
Mi-Ri Gwon (babymil1122@hanmail.net)
Mi-sun Lim (mslim@yu.ac.kr)
Hae Won Lee (haewonbbc@gmail.com)
Young-Ran Yoon (yry@knu.ac.kr)
Dong Heon Yang (ddhyang@knu.ac.kr)
Kwang-Il Kwon (kwon@cnu.ac.kr)
Seunghoon Han (waystolove@catholic.ac.kr)

Version: 6 Date: 8 December 2014

Author’s response to reviews: see over
Cover letter – revised manuscript submission  
DEC08, 2014

Christopher Morrey  
Executive Editor  
BMC Pharmacology and Toxicology

Enclosed is a revised manuscript of an original research article entitled “Population pharmacokinetic and pharmacodynamic modeling of transformed binary effect data of triflusal in healthy Korean male volunteers”. On behalf of my colleagues, I would like to submit this manuscript to BMC Pharmacology and Toxicology.

We appreciate all of your efforts for reviewing this manuscript and all the precious comments you gave us. We carefully considered the comments from you and revised our manuscript. A brief summary of revision outcome is as follows:

1. Unnecessary capitalization was converted to lower case.
2. A concise summary for this study design was added to the end of title.
3. A trial registration number and date of registration were given at the end of the abstract (line 71 – 72).
4. A description of sample size calculation was added (line 288 – 293).
5. A CONSORT flow diagram showing the subject disposition was added to figure 3 (line 320 – 321).

This manuscript was read and approved by all authors.

Thank you for your time in reviewing this submission. If any questions should arise regarding this submission or during the review process, please contact Seunghoon Han, MD, PhD.

Sincerely yours,

Seunghoon Han, MD, PhD

Dept. of Pharmacology, the Catholic Univ. of Korea  
222 Banpodaero, Seocho-gu, Seoul, Korea 137-701  
Phone : +82.2.2258.7326  
waystolove@catholic.ac.kr